Author's response to reviews

Title: "Not just another walking program": Everyday Activity Supports You (EASY) Model - A pilot study for a parallel randomized controlled trial

Authors:

Maureen C. Ashe (Maureen.Ashe@ubc.ca)
Meghan Winters (mwinters@sfu.ca)
Christiane A. Hoppmann (choppmann@psych.ubc.ca)
Martin G. Dawes (martin.dawes@familymed.ubc.ca)
Paul Gardiner (p.gardiner@sph.uq.edu.au)
Lora M. Giangregorio (lora.giangregorio@uwaterloo.ca)
Ken M. Madden (kenneth.madden@ubc.ca)
Megan M. McAllister (Megan.McAllister@hiphealth.ca)
Gillian Wong (ggf.wong@gmail.com)
Joseph H. Puyat (jpuyatubc@gmail.com)
Joel Singer (singerjoel@hotmail.com)
Joanie Sims-Gould (simg@mail.ubc.ca)
Heather A. McKay (heather.mckay@ubc.ca)

Version: 4 Date: 3 November 2014

Author's response to reviews: see over
Dr. Gillian Lancaster  
**Editor-in-Chief**  
Pilot and Feasibility Studies  
c/o BioMed Central  
236 Gray's Inn Road  
London WC1X 8HB  
United Kingdom

**RE:** “Not just another walking program”: Everyday Activity Supports You (EASY) Model – A randomized pilot study for a parallel randomized controlled trial

Dear Dr. Lancaster,

Thank you for the review of our manuscript. The comments and suggestions have significantly strengthened our work. Please find enclosed our revised paper; as requested, we changed the title to better reflect the work that we completed. The title is: “Not just another walking program”: Everyday Activity Supports You (EASY) Model – A *randomized pilot study for a parallel randomized controlled trial*.

We appreciate your consideration of our manuscript in Pilot and Feasibility Studies and look forward to hearing from you.

Sincerely,

Maureen Ashe PhD, PT